Author's response to reviews

Title: Expression of Multidrug Resistance markers ABCB1 (MDR-1/P-gp) and ABCC1 (MRP-1) in Renal Cell Carcinoma

Authors:

naomi Walsh (naomi.walsh@dcu.ie)
annemarie Larkin (annemarie.larkin@dcu.ie)
susan Kennedy (susan.kennedy@rveeh.ie)
liisa Connolly (l.connolly@qub.ac.uk)
jo Ballot (joBallot@ccrt.ie)
wei Ooi (wei.ooi@gmail.com)
giuseppe Gullo (giuseppe.gullo@humanitas.it)
john Crown (john.crown@icorg.ie)
martin Clynes (martin.clynes@dcu.ie)
lorraine O Driscoll (lodrisc@tcd.ie)

Version: 2 Date: 8 June 2009

Author's response to reviews:

Andrea Bucceri,
Senior Assistant Editor,
BMC-series Journals,
Bio Med Central.
8 June 2009.

Re : MS 1743154841261338
“Expression of Multi Drug Resistance markers ABCB1 (MDR-1/ Pgp) and ABCC1 (MRP-1) in Renal Carcinoma”.

Dear Andrea,

Thank you for your email indicating acceptance of our manuscript for publication in BMC Urology. As suggested an authors contribution section has now been added. As suggested by the second round of reviewers we have included the following very useful comments in our revised manuscript.

• P-gp is now spelled consistently throughout manuscript.
• Spelling of Sunitinib corrected.
• A more recent up to-date reference for overall survival in metastatic RCC has been included, the reference of Motzer et al., (1996) has been removed.
• The term “approximately” on P7 has been removed – exact numbers are stated.
• The reference of Filosa & Fabiani has been added to discussion.


• In addition, to above suggestions, a more up to-date reference for Sunitinib/new targeted therapies in RCC has been added.

Yours Sincerely,
Annemarie Larkin.